Author's response to reviews

Title: The effect of continuous ultrasound on chronic non-specific low back pain: A single blind placebo-controlled randomized trial

Authors:

Safoora Ebadi (s_ebadi@razi.tums.ac.ir)
Noureddin Nakhostin Ansari (nakhostin@sina.tums.ac.ir)
Soofia Naghdi (naghdi@sina.tums.ac.ir)
Shohre Jalaeei (jalaeeish@tums.ac.ir)
Mirmostafa Sadat (sadat1335@gmail.com)
Hosein Bagheri (hbagheri@tums.ac.ir)
Maurits W vanTulder (maurits.van.tulder@falw.vu.nl)
Nicholas Henschke (nhenschke@georgeinstitute.org.au)
Ehsan Fallah (fallah_e@razi.tums.ac.ir)

Version: 6 Date: 19 September 2012

Author's response to reviews: see over
Dear Editor,

We are pleased to resubmit the revisions made to the manuscript “The effect of continuous ultrasound on chronic non-specific low back pain: A single blind placebo-controlled randomized trial”

Your comments are fully addressed in the manuscript as below.

Sincerely yours,

Safoora Ebadi

P13, line 3 from bottom change 'deference' to 'difference'

Response:

'deference' was changed to 'difference'

P14, line 4 from bottom change the sentence to read Patients in both groups could have benefitted from the Placebo effect....

Response:

The sentence was changed to “Patients in both groups could have benefitted from the Placebo effects of the treatment”